Author’s response to reviews

Title: Pediatric Ramsay Hunt syndrome: A case report of full recovery with a shorter course of acyclovir and steroid at lower doses

Authors:

Gwinyai Masukume (parturitions@gmail.com)
Sheenah Chibwowa (drsheenah@hotmail.com)
Mbongeni Ndlovu (mbongenigm@yahoo.co.uk)

Version: 6 Date: 11 May 2011

Author’s response to reviews: see over
Dear Editor,

We have removed Figure 1 as per your editorial request.

Sincerely yours,

GM, SC and MN